Author's response to reviews

Title: Delayed diagnosis of spontaneous extrusion of a renal calculus in an adult male with spina bifida and paraplegia: case report. Measures to be taken to reduce medical errors in spinal cord injury patients

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Bakul M Soni (Bakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (gurpreet.singh@ukgateway.net)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 8 Date: 9 Dec 2001

Dear Editor
As advised by you, an acknowledgement section has been included in this revised version to say that the patient gave written consent for publication of clinical photograph.
Thank you for your valuable help
Yours sincerely
Dr S Vaidyanathan